Author's response to reviews

Title: Susceptible Gene of Stasis-Stagnation Constitution from Genome-Wide Association Study Related to Cardiovascular Disturbance and Possible Regulated Traditional Chinese Medicine

Authors:

Kuo-Chin Huang (huangimd@gmail.com)
Hung-Jin Huang (u9947002@cmu.edu.tw)
Ching-Chu Chen (chingchu@ms15.hinet.net)
Chwen-Tzuei Chang (chwentic@mail.cmuh.org.tw)
Tzu-Yuan Wang (yuan.w16@msa.hinet.net)
Rong-Hsing Chen (paishulan@yahoo.com.tw)
Calvin Yu-Chian Chen (ycc929@mit.edu)
Fuu-Jen Tsai (d0704@mail.cmuh.org.tw)

Version: 2 Date: 24 July 2014

Author's response to reviews: see over
Calvin Yu-Chian Chen,
No 2, Yuh-Der Road
Taichung 40447
Taiwan

Dear Editor-in-Chief:

This is a genome-wide association study of Yu-Zhi constitution (one type of traditional Chinese medicine body constitution, indicated stasis and stagnation) in patients with type 2 diabetes and database screen for candidate traditional Chinese medicine ligand by docking and molecular dynamics simulation. There are no any competing financial interests in relation to this work.

Sincerely Yours

Calvin Yu-Chian Chen,
China Medical University Hospital
No 2, Yuh-Der Road, Taichung 40447, Taiwan
Jul 07, 2014
yce929@MIT.edu